ഹ



JGR Earth Surface

RESEARCH ARTICLE

10.1029/2024JF007948

Key Points:

- From 2009 to 2021, salt marshes in Maine have experienced a 16% increase in pool cover
- Mega-pools are larger at lower elevations of the marsh and expand the most when merging with other pools
- Pool expansion can adversely impact a marshes ability to sequester carbon, supporting the need for restoration efforts on Maine's marshes

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K. DeWater, kdewater@une.edu

Citation:

DeWater, K., Kochtitzky, W., Ellis, R., Merrill, P., Pittsley, M., Morgan, P., et al. (2025). Widespread expansion of salt marsh pools observed on Maine marshes since 2009. *Journal of Geophysical Research: Earth Surface*, *130*, e2024JF007948. https://doi.org/10.1029/ 2024JF007948

Received 18 JUL 2024 Accepted 13 FEB 2025

Author Contributions:

Conceptualization: K. DeWater, W. Kochtitzky, P. Morgan Formal analysis: K. DeWater, W. Kochtitzky, C. Burns Funding acquisition: W. Kochtitzky Investigation: K. DeWater Methodology: K. DeWater, W. Kochtitzky Software: K. DeWater, W. Kochtitzky Supervision: W. Kochtitzky, P. Morgan, A. Campbell, S. Adamowicz Validation: R. Ellis, P. Merrill, M. Pittsley Visualization: K. DeWater Writing - original draft: K. DeWater Writing - review & editing: W. Kochtitzky, P. Morgan, C. Burns, A. Campbell, S. Adamowicz

© 2025 The Author(s).

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Widespread Expansion of Salt Marsh Pools Observed on Maine Marshes Since 2009

K. DeWater¹ ^(b), W. Kochtitzky¹ ^(b), R. Ellis¹, P. Merrill¹, M. Pittsley¹, P. Morgan¹, C. Burns² ^(b), A. Campbell^{3,4} ^(b), and S. Adamowicz⁵

¹School of Marine and Environmental Programs, University of New England, Biddeford, ME, USA, ²Chesapeake Bay National Estuarine Research Reserve - MD, Annapolis, MD, USA, ³Goddard Space Flight Center, NASA, Rockville, MD, USA, ⁴Goddard Earth Sciences Technology and Research II, University of Maryland, Baltimore County, MD, USA, ⁵Rachel Carson National Wildlife Refuge, Wells, ME, USA

Abstract Salt marshes provide critical habitats, protect coastal infrastructure, and are increasingly exposed to sea level rise, with many having a history of agricultural use and ditching over the centuries. Pool area coverage can be considered an indicator of marsh health but is rarely quantified. In this study, we digitized marsh pools using aerial imagery to quantify changes in pool area and density on 12 salt marshes in Maine from 2009 to 2021 as a case study of marsh response to environmental changes. We categorized pools into three types: mega-pools, individual pools, and perimeter pools, based on morphology and examined whether pools remained singular, split, or combined. We found a 15.7% increase in pool area from 2009 to 2021 on all marshes, primarily driven by mega-pool expansion, whereas individual and perimeter pools remained relatively constant. The rate of pool expansion across all marshes was 49,000 m² a⁻¹ with mean mega-pool size 6,400 \pm 400 m². There was an increase in pool count per km² on all marshes except for the York River marsh, which still experienced area expansion. Pools primarily increase in cover through merging or being engulfed by mega-pools. Area cover change was not substantial when pools remained singular, split into many pools, or were only present in 2009 or 2021. Mega-pools were larger on lower marsh elevations and expanded at a greater rate when overlapping ditches, suggesting influence by sea level rise and historic agriculture. Marsh restoration projects that promote the drainage and re-vegetation of mega-pools may reverse this trend.

Plain Language Summary Salt marshes are habitats for a variety of species, many of which are endangered or threatened. Marshes also provide a buffer to coastal infrastructure and reduce damage from coastal flooding. Continued expansion of standing water, or pools (also called ponds), on marshes could cause habitat loss and limit their ability to perform ecosystem services. To quantify pool change, we mapped salt marsh pools present in 2009 and 2021 for 12 salt marshes in Maine. Maine's salt marshes experienced notable pool development, having increased standing water on marsh surfaces and reduced grass cover. We grouped pools based on shape: mega-pool (large, disconnected pools, associated with expansion), individual pools (small, singular pools, defined borders), and perimeter pools (long, thin pools, along embankments). We found an increase in pool area from 2009 to 2021 in all marshes driven by mega-pool expansion. The primary way in which pool cover increased was through pools combining to form mega-pools or being overtaken by existing mega-pools. Expansion could be occurring because of increased flooding due to sea level rise, prompting erosion, and lasting effects of historic agricultural practices on the marsh surface. Future marsh restoration projects promoting the drainage and re-vegetation of mega-pools may reverse this trend.

1. Introduction

1.1. Marshes, Sea Level Rise, and Human Impacts

Salt marshes are dynamic ecosystems that buffer coastlines from intense storms (Temmerman et al., 2012), sequester blue carbon (Brewer et al., 2023), and provide critical habitat for endangered and rare species (Wigand et al., 2017). Northeast U.S. salt marshes have experienced degradation due to human activities such as agricultural drainage, tidal restrictions, and ditching (Adamowicz et al., 2020; Burdick et al., 2020; J. A. M. Smith et al., 2022), along with climate change-driven sea level rise (Hartig et al., 2002; Kearney et al., 1988; Watson et al., 2016). In the 19th and 20th centuries, ditches were constructed for mosquito control, and farmers used ditches, dikes, and embankments to drain sections of marshes for hay cultivation, which now has lasting effects on pool development (Burdick et al., 2020; Sebold, 1998; D. C. Smith & Bridges, 1982; D. C. Smith et al., 1989). Sea



level rise contributes to increased inundation and changes in tidal dynamics, while historic ditching disrupts natural drainage patterns and can lead to altered flow and sediment deposition (Corman et al., 2012; J. A. M. Smith et al., 2022; S. M. Smith et al., 2017). Hydrologic alteration can prompt the transition of vegetated marsh to pools and mudflats as marshes erode due to frequent and sustained inundation (Schepers et al., 2020). Pool expansion and formation may be an indication of increased inundation, though pools may also result from other mechanisms (Cavorta et al., 2003; Kearney et al., 1988). Given these environmental changes, Northeast salt marshes are an exemplary location to better understand changing morphology and marsh response. There is a growing need to document pool changes to support marsh restoration and to monitor the efficacy of future restoration projects globally (McKown et al., 2023; Watson et al., 2016; Wigand et al., 2017).

Threshold sea level rise rates for marsh survival depend on sediment supply, tide range and marsh inundation, and site-specific vegetation. Studies throughout the Northeast have indicated marsh vulnerability to submergence due to accelerated sea level rise using measurements and modeling (Raposa et al., 2017; Watson et al., 2017). For Maine's marshes, vertical accretion rates were measured to be 2.8 mm a^{-1} from 1986 to 2003, and most marshes could keep pace with sea level rise at that time (Goodman et al., 2007). However, accretion rates may vary among locations within a marsh, particularly high versus low marsh, at a single location within a marsh over time, and between nearby marshes (Yellen et al., 2022). Therefore, it is difficult to draw conclusions about marsh response to sea level rise based on accretion rates. However, with sea levels rising at 3 mm a^{-1} since 1990 in Portland, Maine and accelerating (Fernandez et al., 2020), Maine marshes may not be keeping pace, leading to inundation and vegetation dieback. Here, we provide observational evidence that thresholds may have been crossed for many of Maine's marshes by quantifying pool expansion, which can be indicative of increased inundation (Cavorta et al., 2003).

1.2. Salt Marsh Pools

The underlying mechanism for new pool formation is initial peat collapse due to the loss of living root network structures with subsequent pool expansion through erosion (Chambers et al., 2019; DeLaune et al., 1994; Himmelstein et al., 2021). Root networks are lost due to vegetation dieback, which can be caused by wrack deposition, ice rafting, and inundation from changing drainage channels (Erwin, 1996; Pethick, 1974; C. A. Wilson et al., 2014). Sustained inundation can also be caused by sea level rise (Temmerman et al., 2012) or agricultural embankments interfering with hydrology (Adamowicz et al., 2020; Burdick et al., 2020), making marsh grasses such as *Spartina alterniflora* and *Spartina patens* increasingly vulnerable to mortality due to saline stress, low oxygen, and high sulfide exposure (Alber et al., 2008; DeLaune et al., 1994; Kirwan & Guntenspergen, 2012). Pools can be evidence of a history of agriculture in marsh systems, as typified on the Webhannet marsh in Maine (Adamowicz et al., 2020; K. R. Wilson et al., 2010). It has been supported that most pools are secondary in origin or formed after the development of the marsh platform due to disturbances (K. R. Wilson et al., 2010). Due to the variety of disturbances to marsh surfaces, including the longstanding legacy of agriculture, it is difficult to discern the origin of existing pools.

Pools have been observed to form and drain to maintain an overall stable area cover on marshes in dynamic equilibrium. After a pool has formed, branches can develop from the main tidal channel to drain the pool, reducing inundation and saline stress and allowing for subsequent vegetation recovery (K. R. Wilson et al., 2009, 2014). However, it has more recently been shown that pools can undergo continual erosion following connection to drainage without vegetation recovery. Continued erosion after drainage is attributed to increased inundation from sea level rise and the exposure of soft-substrate pool bottoms to wave and tide action (Himmelstein et al., 2021; Mariotti et al., 2020). Unvegetated soil exposure can lead to widened drainage channels and runaway pool expansion, resulting in a loss of vegetated marsh surface and sediment export (Burns, Alber, & Alexander, 2021; Burns, Alexander, & Alber, 2021; Schepers et al., 2020). On intermediate and large sized pools, expansion is also caused by wind wave-driven edge erosion, whereas erosion of smaller pools is driven by biogeochemical processes (Himmelstein et al., 2021; Mariotti & Fagherazzi, 2013). As vegetation dieback occurs without recovery and pools expand, there may be a phase shift between the alternative stable states of vegetated marsh to unvegetated mudflat (Himmelstein et al., 2021; Kirwan & Murray, 2007; Morris et al., 2002).

The surface layer of vegetated marsh soils is notably efficient at sequestering carbon from the atmosphere, and states in New England have estimated the amount of carbon stored in marshes to count toward carbon neutrality goals (Alongi, 2020; Brewer et al., 2023). Though soil characteristics unique to pools are shallow and do not



extend deep into the marsh vertically, it has been shown that soil surrounding expanding pools and inundated pannes store significantly less carbon and have decreased sediment stability compared to soils surrounding pools that are not expanding (Berkowitz et al., 2018; Kirwan & Mudd, 2012; Luk et al., 2023; Vincent et al., 2013, 2014). Pool expansion and development can shift the carbon balance of the marsh to be a carbon source instead of a sink as vegetation loss and sediment export are accelerated (Chambers et al., 2019; Himmelstein et al., 2021; Schepers et al., 2020). Therefore, understanding the patterns of changing pools on Maine marshes may help to inform carbon estimates and goals.

On the Webhannet marsh in Maine, there was an overall decrease in pool area from 1962 to 2003 (K. R. Wilson et al., 2009). However, pool development and expansion has since been anecdotally noted on the Webhannet, Biddeford Pool, and Goosefare Brook marshes in Maine and provide the basis for our analysis here. In Massa-chusetts, this pool expansion was described as mega-pooling, or runaway pool development in which pools continually expand (McKown et al., 2023).

Restoration efforts to develop shallow runnels to connect these mega-pools to drainage and promote re-vegetation have been effective (McKown et al., 2023) but may not be a long-term solution to assist marshes in adapting to sea level rise (Besterman et al., 2022). Additionally, the ability of marshes to sequester carbon and remain carbon sinks may be hindered by pool development that leaves soil exposed to erosion and decreases vegetation cover (Brewer et al., 2023). Therefore, it is necessary to monitor changes in marsh pool dynamics for the management and evaluation of future restoration efforts. Pool coverage has yet to be quantified on many of New England's salt marshes, including those in Maine, which is fundamental to understanding the impacts of sea level rise and historic agricultural practices on marsh hydrology and overall resilience.

1.3. Study Objectives

In this study, we aimed to understand how salt marsh pools have changed in area based on morphology, elevation, and their coincidence with ditches. We focused on 12 of the largest salt marshes in Maine (Figure 1), using them as a case study to understand pool dynamics and changes in the region. We hypothesized that there would be an increase in pool area, particularly of mega-pool area, across the 12-year period on all marshes. Therefore, the purpose of this research is to

- 1. Quantify changes in pool size, pool count, and pool type from 2009 to 2021
- 2. Categorize pools based on morphologies and expansion patterns (i.e., Mega-pools)
- 3. Compare pool changes to marsh elevation and relative rates of sea level rise
- 4. Generate a census of digitized pools for the largest salt marshes in southern Maine as a baseline for future restoration projects and impacts of climate change
- 5. Inform management decisions about which marshes need to be prioritized for restoration

2. Materials and Methods

2.1. Imagery

We selected every marsh described as "salt or brackish marsh" with *Spartina* as the dominant community in Southern Maine larger than 0.5 km² as classified by the Maine Natural Areas Program Current Tidal Marshes layer (Gawler & Cutko, 2015; Odum, 1988). The 12 marshes in Maine that were selected (in order of decreasing area) were Scarborough, Webhannet River, Spurwink, Marshall Point, York River, Ogunquit River and Stephens Harbor, Brave Boat Harbor, Cousins River, Biddeford Pool, Little River, Goosefare Brook, and Gooch's Beach marshes (Figure 1; see Table S2 and S3 for site descriptions). Ten of the marshes are back barrier marshes, separated from the coastline by sandy beaches and dunes; the York River and Cousins River marshes are along tidal rivers (Kelley et al., 1988).

We downloaded aerial photos from the National Agricultural Imagery Program (NAIP) through the United States Geological Survey Earth Explorer for 2009 and 2021. NAIP provides high spatial resolution and open-access imagery every four years, which is necessary for sufficiently accurate digitization. We selected the year 2021 because it is the most recent imagery available. The year 2009 was chosen because it is the earliest imagery available with RGB values, ensuring consistency in digitization by using color imagery rather than black and white. All aerial photos for 2009 were collected in July, and the 2021 aerial photos were taken in July and October (Tables S1–S3). Although some marshes were compared between summer and fall due to data availability, our





Figure 1. Map of the selected marshes for which pools were digitized. Marshes are shown in green. The marsh polygons were retrieved from the Maine Natural Areas Program Current Tidal Marshes layer.

data set aims to capture long-term trends rather than short-term seasonal cycles. The spatial resolution of the aerial photos is 1 m for 2009 and 0.6 m for 2021. All aerial photos were taken during comparable low tidal stages and were not collected during periods of extreme precipitation events (Table S2 and S3).

2.2. Digitization

We define salt marsh pools following Adamowicz & Roman (2005)as depressions on a marsh that retain water throughout a tidal cycle and are not connected to tidal channels. Salt marsh pools as we define them are also referred to as ponds (Schepers et al., 2020) or interior pools (Watson et al., 2016). For this study, we did not include pannes, defined as depressions on the marsh surface that do not retain standing water (Adamowicz & Roman, 2005), since they have not been linked to the feedback loop of salt marsh erosion (Schepers et al., 2020). We further defined pools into three categories as their features were apparent and consistent in all observed marshes (Figure 2):

• Mega-pool: Defined as disjointed borders, typically in higher marsh, and characterized by being surrounded by smaller fragmented pools (described by Himmelstein et al. (2021); named by McKown et al. (2023)), typically have areas greater than hundreds of square meters.





Figure 2. Examples of the three pool types. (a) Mega-pool on the Biddeford Pool marsh from a nadir view; (b) the same megapool from an oblique view; (c) individual pool on the Little River marsh from a nadir view; (d) the same individual pool from an oblique view; (e) perimeter pool on the Marshall Point marsh from a nadir view; (f) the same perimeter pool from an oblique view. Nadir images on the left show pools as they appear in 5 July 2009 NAIP imagery; oblique images on the right were collected with a drone on 23 March 2024.

- Individual pool: Defined borders, typically circular and primarily naturally occurring (described by Redfield (1972), Adamowicz and Roman (2005), and McKown et al. (2023)), typically have areas less than hundreds of square meters.
- Perimeter pool: Defined as disjointed borders, long, thin pools noted for formation along embankments on a marsh's border or roads

Stream digitizing was used in conjunction with a One by Wacom tablet in QGIS version 3.28.0 to manually polygonize the extent of salt marsh pools and pool systems (following K. R. Wilson et al. (2009), Watson et al. (2016), Burns, Alber, and Alexander (2021), and Burns, Alexander, and Alber (2021)). While digitizing, pools were given an attribute of either mega-pool, individual pool, or perimeter pool. Mega-pools and perimeter

pools were delineated by the extent of their disjointed edges, which in some cases resulted in small areas of vegetation being included within the pool boundaries. However, our goal was to identify the full extent of the marsh surface encompassed by the mega-pools; therefore, the edge of each pool polygon was placed at the outermost extent of standing water associated with each pool. All pools were digitized on a scale of 1:400 to 1:1000 in UTM 19N.

After pools were digitized on all marshes for both years, each pool was attributed a unique identification based on the pool type (i.e., mega, individual, perimeter). Based on this unique identification, we used spatial joins and geoprocessing intersection tools to identify changes in pools categorized as: remained, combined, split, absent in 2021, or formed before 2021. Indexing pools by morphology and change type allowed us to identify pools that remained singular in both years, pools that merged, pools that separated, pools only present in 2009, and pools only present in 2021.

Imagery from 2009 had poor location accuracy, which caused georeferencing errors between digitized pools in 2009 and 2021. While this discrepancy was not prevalent throughout the data set, and offset pools were typically within 2 m of their correct location, it still affected less than 850 small individual pools, each measuring less than 5 m wide. As a result, we were unable to spatially join the two data sets for these pools and track them across our study. To correct this error, all pools with georeferencing errors were identified and manually moved to overlap with the 2021 air photos. We therefore provide two different layers for 2009, one with the corrections to make the layers internally consistent with 2021 and one without the correction to align with the 2009 air photos. The area of the pools is the same, but the position of the pool differs.

2.3. Quantification

After georeferencing errors were corrected, pool size, percent cover, and type were analyzed, and pool areas were calculated. Each marsh area as defined by the Maine Natural Areas Program Current Tidal Marshes layer was calculated. These calculations were used to determine pool cover, type, and changes between 2009 and 2021. Using intersections, we identified which pools overlapped with ditches in the digitized ditch layer from the U.S. Geological Survey and compared changes in pool area based on these intersections (Peck et al., 2024).

2.4. Elevation Analysis

For this analysis, we used the best available digital elevation model (DEM) from the United States Geological Survey (OCM Partners, 2024) derived from 2020 lidar data, downloaded from the NOAA Data Access Viewer with a vertical datum of NAVD88. The DEM has a reported vertical accuracy of 4.21 cm and a horizontal accuracy of 0.37 cm, with 0.7 m spatial resolution. The DEM was processed using custom python scripts to calculate the mean marsh elevation within the marsh polygons provided by the Maine Natural Areas Program Current Tidal Marshes Layer (DeWater et al., 2024). The mean elevation of each pool was calculated by determining the mean elevation of the marsh within the digitized pool polygons using the DEM. Because the DEM was hydro-flattened during processing, the elevation of pools appears flat; therefore, the calculated mean elevation for each pool represents the elevation at which the pool would be flooded by tides (OCM Partners, 2024). Each pool was then categorized into tidal datum bins: Lowest Astronomical Tide (LAT), Mean Lower Low Water (MLLW), Mean Low Water (MLW), Mean High Water (MHW), Mean Higher High Water (MHHW), or Highest Astronomical Tide (HAT). For example, a pool in the MHW bin would be flooded by tides at that level and higher but not by tides less than MHW. The tidal datum values were determined in NAVD88 based on NOAA's reported tidal data for the Portland tide gauge. Differences in tidal datums across our survey region fall within the stated uncertainty of conversion between NAVD88 and tidal datums along with DEM uncertainty, so we rely on the Portland datum conversion to identify regional trends.

2.5. Uncertainty Analysis

To minimize inconsistencies from digitizer bias, only one person digitized all 24,712 pools in this study. All digitized layers were checked by at least two independent reviewers for accuracy of digitization and completeness. Layers were reviewed for potential discrepancies caused by tides and shadows. Any pool covered by a tide or shadowed in 1 year but not the other was removed from the data set. This was infrequent, occurring fewer than 50 times. Because pools are difficult to discretely define, our results have better relative than absolute accuracy.

Table 1

Marsh Area, Pool Area for 2009 and 2021, and Number of Pools in 2009 and 2021

Name	Marsh area (km ²)	Pool area 2009 (m ²)	Pool area 2021 (m ²)	Number of pools 2009	Number of pools 2021
Scarborough	10.1	$1,615,000 \pm 9,100$	$1,900,000 \pm 6,700$	4,812	7,345
Webhannet River	4.88	$689,200 \pm 5,400$	$727,800 \pm 3,500$	2,719	2,970
Spurwink	1.91	$265,800 \pm 3,400$	$302,900 \pm 2,400$	950	1,064
Marshall Point	1.81	$250,700 \pm 3,600$	$273,900 \pm 2,500$	439	582
York River	1.59	$201,100 \pm 2,700$	$232,600 \pm 2,000$	82	75
Ogunquit and Stephens Harbor	1.31	$220,300 \pm 2,800$	$277,800 \pm 2,100$	566	678
Brave Boat Harbor	1.06	$91,700 \pm 1,700$	$115,600 \pm 1,300$	196	297
Cousins River	0.86	$54,200 \pm 1,400$	$71,500 \pm 1,000$	155	218
Biddeford Pool	0.81	$63,300 \pm 1,400$	$75,200 \pm 980$	85	123
Little River	0.80	$186,600 \pm 2,800$	$204,900 \pm 1,900$	355	507
Goosefare Brook	0.71	$68,600 \pm 2,200$	$112,800 \pm 1,300$	115	213
Gooch's Beach	0.44	$41,100 \pm 1,400$	$50,600 \pm 870$	82	84
Total	27.3	$3,748,000 \pm 13,200$	$4,335,000 \pm 9,400$	10,556	14,156

Note. Pool area is provided as pool area \pm uncertainty from Equation 1.

We considered digitizer accuracy (D) and pixel resolution (P) as our primary sources of uncertainty and the only uncertainties we can quantify (Burns, Alber, & Alexander, 2021; Burns, Alexander, & Alber, 2021; Crowell et al., 1991; Fletcher et al., 2003). Other sources of uncertainty could include environmental variables (i.e., lighting, tide stage, rain, etc.), which we note for each image/marsh (Table S2). Seasonal variations in pools are also a source of error but are unable to be quantified due to a lack of data availability (Tables S1–S3). Digitizer accuracy depends on the digitizer's interpretation of varying imagery and their ability to consistently identify pool extents. Digitizer accuracy may vary based on pool type as individual pools, typically smaller than mega-pools, were easier to digitize than larger mega-pools. To quantify this error, digitizer consistency was calculated for each time frame of imagery and for each pool type (mega, individual, and perimeter pool). Three mega, individual, and perimeter pools were digitized 10 times for 2009 and 2021. The standard deviation of area was used to quantify digitizer accuracy by pool type and year. Uncertainty due to pixel (spatial) resolution of the aerial imagery impacted the data set in that it determined the accuracy of pool extent delineation. Only pools greater than the pixel size could be identified and digitized. These two uncertainties can be expressed as a single value of uncertainty (U) since they are random and uncorrelated. This uncertainty value is the square root of the sum of both error sources (Equation 1) (Burns, Alber, & Alexander, 2021; Burns, Alexander, & Alber, 2021; Crowell et al., 1991; Fletcher et al., 2003).

$$U = \pm \sqrt{D^2 + P^2} \tag{1}$$

3. Results

Pool area increased on all 12 marshes from a total of $3,748,000 \pm 13,200 \text{ m}^2$ to $4,335,000 \pm 9,400 \text{ m}^2$ between 2009 and 2021 at an overall rate of 49,000 m² a⁻¹. The number of pools on all marshes increased from 10,566 to 14,156 between 2009 and 2021 at an overall rate of 300 pools a⁻¹ (Table 1). The mean pool size (±Uncertainty) across all marshes in 2021 for mega-pools was $6,400 \pm 400 \text{ m}^2$; individual pools was $50 \pm 27 \text{ m}^2$; and perimeter pools was $300 \pm 100 \text{ m}^2$. There is no notable trend in the data between marsh size, marsh location (north vs. south), or type of marsh (back barrier vs. along a tidal river).

Little River marsh had the greatest pool area cover at $23.62\% \pm 0.01\%$, followed by Ogunquit marsh and Scarborough marsh at $21.18\% \pm 0.01\%$ and $18.75\% \pm 0.01\%$, respectively, in 2021. The Cousins River marsh had the least pool area cover at $8.34\% \pm 0.01\%$ in 2021. On all marshes, an increase in pool percent cover was observed between 2009 and 2021 with the average increase being $+18\% \pm 7\%$. The largest increase in pool percent cover occurred on Goosefare Brook marsh at $6.18\% \pm 0.01\%$. Ogunquit marsh had the second greatest



10.1029/2024JF007948



Figure 3. Percent cover of pools (%) on each marsh in 2009 and 2021. Bars are separated into the proportion of mega-pools in dark gray, individual pools in light gray, and perimeter pools in white. Uncertainty error bars represent the thickness of the bar outlines.

change in percent cover at $4.38\% \pm 0.01\%$. The smallest increase in pool percent cover was observed on Webhannet marsh at $0.76\% \pm 0.01\%$ (Figure 3).

Most pool expansion was driven by an increase in mega-pool area cover. Goosefare Brook marsh experienced a $6.01\% \pm 0.01\%$ increase in mega-pool cover with a $-0.032\% \pm 0.01\%$ change in individual pool cover from 2009 to 2021 and only a $0.21\% \pm 0.01\%$ increase in perimeter pool cover. On the Biddeford Pool marsh, there was a $1.35\% \pm 0.01\%$ increase in mega-pool cover with only a $0.06\% \pm 0.01\%$ increase for both individual pools and perimeter pools. A $2.33\% \pm 0.01\%$ increase in mega-pool cover occurred on the Scarborough marsh with a $0.36\% \pm 0.01\%$ and a $0.04\% \pm 0.01\%$ increase in individual pool and perimeter pool cover, respectively (Figure 3).

On all marshes, there was an increase in mega-pool density (Figure 4). Little River and Ogunquit marshes show the greatest density of mega-pools at 47 mega-pools per km^2 and 31 mega-pools per km^2 , respectively, in 2021. The greatest increase in pool density was observed on Goosefare Brook marsh from 18 mega-pools per km^2 in 2009 to 28 mega-pools per km^2 in 2021. The York River marsh experienced a decrease in total pool count per km^2 with a decrease in the number of individual pools but an increase in the number and area cover of mega-pools (Table 1, Figures 3 and 4).

The majority of the observed increase in pool cover occurred when pools combined (Figure 5). Pools that were only present in 2009 or 2021 and pools that remained, were present and remained singular in both years, did not substantially contribute to the increase in area cover. Pools that split from being singular in 2009 to many pools in 2021 experienced an overall decrease in area cover; Biddeford Pool marsh was an exception, however, with a 0.5% increase in area of pools that split (Figure S1 in Supporting Information S1). Goosefare Brook marsh experienced the greatest increase in pool cover (Figure 5), driven by pools combining, which had over a 1,000% increase in area (Figure S1 in Supporting Information S1). Goosefare Brook was followed by the Scarborough and Little River marshes, which had a 600% and 300% increase in the area of combined pools, respectively (Figure S1 in Supporting Information S1). Brave Boat, Goosefare, Ogunquit, and Cousins River marshes experienced the greatest increase in the area of remaining pools at over 20% increases (Figure 5; Figure S1 in Supporting Information S1).



10.1029/2024JF007948



Figure 4. Pool density (number of pools per km^2) on each marsh in 2009 and 2021. Bars are broken into the number of megapools, individual pools, and perimeter pools per km^2 .

Marsh pools changed in a variety of ways from 2009 to 2021. Mega-pool expansion was prevalent across all marshes, commonly combining with individual pools (Figure 3; Figures 6a–6d). In some instances, individual mega-pools decreased in size from 2009 to 2021 due to shoreline erosion or connection to drainage channels (Figures 6b and 6e). Observed changes in individual pools include being engulfed by mega-pools (Figures 6a–6d) and re-vegetation, though infrequent (Figure 6f). Changes in perimeter pools were infrequent and minimal across all marshes.

All marsh pools occurred above the mean low water elevation but within the highest astronomical tide range, aside from less than 10 pools being above highest astronomical tide across all 12 marshes in 2009 and 2021. The greatest pool expansion was observed in pools between mean higher high water and highest astronomical tide at an increase of $21\% \pm 1\%$. Pools between mean high water and mean higher high water expanded at $8\% \pm 0.6\%$, and pools from mean low water to mean high water expanded at $13\% \pm 1\%$.

4. Discussion

4.1. Widespread Pool Expansion

Salt marshes are essential ecosystems that serve as habitats for many endangered and rare species and protect the coastline from intense storms and flooding. The expansion of marsh pools and subsequent erosion can reduce the marshes' ability to provide these ecological services. Marshes across the Northeast are facing challenging conditions due to the impacts of ditching and agriculture over the last several centuries (Adamowicz et al., 2020) and current rapid sea level rise (Maine Climate Council, 2020; Raposa et al., 2017). Although some evidence suggests that salt marshes can respond to sea level rise by increasing the rate of drainage channel formation and vertical accretion (C. A. Wilson et al., 2014), our results indicate that Maine's marshes are experiencing rapid expansion of mega-pools. This trend of expansion is consistent with observations across the Northeast and could indicate a diminished capacity of marshes to adapt to rising sea level and hinder their ability to provide important ecological functions (McKown et al., 2023; Raposa et al., 2017).

Marsh accretion and ability to adapt to sea level rise depend on various factors such as sediment availability, deposition, vegetation productivity, and nutrient input from surrounding watersheds. The continued expansion of





Figure 5. Percent cover (%) of pools categorized by type of change on each marsh in 2009 and 2021. Present in 2009 and 2021 represents pools that existed in 2009 and 2021 and did not split or merge with any other pools. Merge refers to pools that were separate in 2009 but combined into one pool by 2021. Split refers to pools that were one pool in 2009 but divided into more than one pool by 2021. Only Present 2009 refers to pools that existed in 2009 but not in 2021. Only Present 2021 refers to pools that existed in 2009 but divided into more that did not exist in 2009 but did exist in 2021 (See K. R. Wilson et al., 2009). 2009 bars represent percent cover of pools as they existed in 2009; for example, merge in 2009 represents separate pools before they combined. 2021 bars represent percent cover of pools as they existed in 2021; for example, merge in 2021 represents singular pools after they combined. Uncertainty error bars are less than the thickness of the bar outlines.

pools may interact with these factors and compound marsh instability by promoting vegetation die-off (Himmelstein et al., 2021; Kirwan & Murray, 2007; Kirwan et al., 2010). As vegetation declines, there is less potential for sediment deposition to build up the marsh surface, and soils become more vulnerable to erosion and peat collapse (DeLaune et al., 1994). This could lead to a feedback loop of pool expansion, erosion, and marsh degradation, further limiting the marsh's ability to accrete and keep pace with rising sea levels (Schepers et al., 2020).

While some mega-pools or parts of mega-pools may connect to drainage channels (Figure 6b), they can remain un-vegetated, and their soft-substrate bottoms remain exposed to tidal erosion (Schepers et al., 2020). We observed continued mega-pool expansion despite connection to drainage on these marshes (Figure 6b). If marsh pools were forming, connecting to drainage, and re-vegetating in dynamic equilibrium as suggested by C. A. Wilson et al. (2014), we would expect to see minimal change in pool cover. A minimal change in pool cover was observed with individual and perimeter pools, and on some marshes, there was a decrease in individual pool cover from 2009 to 2021 (Figure 3). However, with mega-pools, there was significant expansion on all marshes, establishing the mega-pools as the primary contributor to the observed increased pool cover (Figure 3).

K. R. Wilson et al. (2009) digitized a sample of 50 pools on the Webhannet marsh in 1962, 1977, 1991, and 2003. They found that most of the pools decreased in area over the 41-year period but that changes were dynamic, with pools increasing and decreasing in area over time. During that time frame, from 1962 to 2003, the rate of relative sea level rise, calculated from monthly mean sea level records was ~1.8 mm a⁻¹ (Fernandez et al., 2020). In contrast, we found an increase in the area of pool cover and the number of mega-pools for all 12 marshes, including the Webhannet marsh, even though our study only covers a 12-year time period. Since 1990, the rate of relative sea level rise observed by the Portland, Maine tide gauge has been 3.05 mm a⁻¹, an increase from 1.8 mm a⁻¹ since 1912 (Fernandez et al., 2020). Although data from the Portland tide gauge suggests minimal change in



10.1029/2024JF007948



Figure 6. Examples of changes in salt marsh pools from 2009 to 2021. In all maps, the north is oriented upwards. Blue polygons outline the extent of mega-pools, green outlines individual pools, and purple outlines perimeter pools. (a) Scarborough marsh, where a mega-pool expanded and engulfed individual pools. (b) Webhannet marsh, where a portion of the mega-pool was connected to channel drainage but also expanded and engulfed individual pools. (c) Spurwink marsh, where a mega-pool expanded along Sawyer Road and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Goosefare Brook marsh, where a mega-pool expanded and engulfed individual pools. (d) Brook marsh, where a mega-pool due to shoreline overwash. (f) Cousins River marsh, where individual pools re-vegetated.

sea level over the 12-year period of this study, due to its coincidence with the 18.5-year nodal cycle, marshes are likely responding to the decades of sea level rise that occurred prior to this specific period.

There are various ways in which pools can change over time (present in 2009 and 2021, split, merge, only present in 2009, only present in 2021 (Figure 5)). For example, the York River marsh was the only marsh in which we documented a decrease in pool density; however, we still observed an increase in pool area, suggesting pools merged and expanded (Figure 4; Table 1). On the Webhannet marsh, K. R. Wilson et al. (2009) documented 60% of pools in what they refer to as the "life" stage, or pools that remained singular, over four-time frames from 1962 to 2003 but did not quantify changes in the area of pools that correspond to different types of changes. Here, we quantified area changes of pools grouped by the type of change, and we refer to their life stage as present in 2009 and 2021. We found that pools present in 2009 and 2021 (remained singular) did not substantially undergo expansion (Figure 5). While our study only examined changes over a 12-year period, we found that pool merging is by far the main driver of pool expansion, whereas splitting often resulted in a decrease in area cover (Figure 5; Figure S1 in Supporting Information S1). Pools formed due to inundation typically develop in depressions on the marsh from past agricultural practices or because they are blocked by a tidal restriction. In these depressions or above tidal restrictions, water collects and inundates the surrounding vegetation, leading to die-off and erosion. Once one pool forms, due to this inundation, it is likely that other pools will form and combine as the inundation continues. Pool splitting suggests deposition of sediment within the pools, thereby increasing the elevation in the pool and lowering the extent of inundation. The formation of new pools was not a notable contributor to the increase in pool area because formation was infrequent as most depressions or areas above tidal restrictions on the marsh surfaces already had pools present. While pools only existed in 2009 and pools splitting led to a decrease in pool area, the decrease was not sufficient to balance the expansion caused by pools merging.



Northeast marshes have a longstanding history of agricultural use and ditching, which have interfered with marsh hydrology patterns (Sebold, 1998; D. C. Smith & Bridges, 1982; D. C. Smith et al., 1989). Agricultural embankments on the marshes can prevent adequate drainage and ditches can cause subsidence, promoting the continued expansion of mega-pools as vegetation dieback occurs from inundation (Adamowicz et al., 2020; Burdick et al., 2020). We observed the presence and formation of perimeter pools along marsh borders adjacent to embankments, occurring infrequently on 10 of the 12 marshes and not at all on Cousins River and Gooch's Beach marshes. It is difficult to quantify the extent to which these underlying ditches and embankments impact the development of mega-pools due to the limited documentation of agriculture in Maine marshes that occurred since the 19th century (Burdick et al., 2020; D. C. Smith & Bridges, 1982; D. C. Smith et al., 1989). However, based on our analysis quantifying pools that intersect with the USGS marsh ditches layer (Peck et al., 2024), we found a greater rate of expansion of pools that coincide with ditches. Of pools that coincide with ditches, area cover increased by $30\% \pm 1\%$, whereas pools that do not coincide with ditches only increased in total area cover by $8\% \pm 0.5\%$. Though we find greater expansion with pools intersecting ditches, this could be explained by ditches preferentially occurring in higher marsh areas, which coincides with mega-pool development.

Mega-pool and perimeter pools were frequently observed forming along roads and railroads, some of which act as tidal restrictions by intersecting channels. Pool formation along roads occurred notably along Sawyer Rd in the Spurwink Marsh (Figure 6c; Figure S2a in Supporting Information S1) and Harbor Rd (previously named Lower Landing Rd) in the Webhannet marsh (Figure S2b in Supporting Information S1) (See Figure S2 in Supporting Information S1) for more examples of pool expansion along roads). Pool formation along embankments was documented along Harbor Rd on the Webhannet marsh from 1962 to 2003 (K. R. Wilson et al., 2009). We identified the pools reported by K. R. Wilson et al. (2009) along Harbor Rd and categorized them as mega-pools which expanded from 2009 to 2021. On the Spurwink marsh, there are plans to remove Sawyer Rd in the coming years because it is frequently flooded by high tides. Removing Sawyer Rd could restore the hydrology of the marsh similar to restoration strategies for correcting drainage (McKown et al., 2023) and allow the mega-pool to drain and re-vegetate. The response of the marsh to road removal should be monitored closely to determine if road and tidal restriction removal could be a potential strategy in improving marsh resilience through restored marsh hydrology.

To understand the impact of elevation on pool size and distribution, we examined the elevation of pools compared with overall marsh elevation. The Goosefare Brook marsh, which by far, had the greatest increase in pool percent cover at 64%, was the only marsh where the mean pool elevation was lower than the mean marsh elevation. Increased exposure to rising sea levels over the past decades likely contributes to the formation and expansion of pools on the marsh, as demonstrated by Goosefare Brook, in which the average pool is lower than the mean marsh elevation and is therefore more exposed to erosive forces from flooding. This may suggest that pool development is impacted by the frequency of inundation and flooding because pools at lower elevations are more often exposed to tides and rising sea levels.

We further found a moderate negative correlation ($R^2 = -0.2$) between pool elevation and mean mega-pool size, indicating that larger mega-pools tended to develop at lower elevations, whereas smaller mega-pools occurred at higher marsh elevations. An opposite trend was observed for individual pools in which there was a moderate positive correlation ($R^2 = 0.4$) between pool elevation and mean individual pool size, suggesting that higher elevations typically have larger individual pools, while lower elevations have smaller individual pools. Similarly, perimeter pools were found to be larger and more frequent at higher marsh elevations. Pools within the tide range of mean higher high water to highest astronomical tide experienced the greatest rate of expansion compared to pools between mean low water and mean high water, as well as pools between mean high water and mean higher high water. With previous sea level rise, lower elevations experienced more frequent and longer duration periods of inundation, where we found mega-pools to be larger in area and are likely still adjusting to higher sea levels. The greater expansion rate of pools in higher tide ranges may further indicate impacts of human modifications of the marsh but is in part confounded by elevation. Historic ditches, which are typically found in higher marshes, can cause elevation depressions, and we found a greater rate of expansion of pools intersecting these ditches, similar to the greater expansion of mega-pools found in higher tide ranges.

4.2. Sediment Erosion

Previously, marsh accretion rates have been compared to rates of sea level rise to predict marsh survival in Maine (Goodman et al., 2007); however, these measures can be difficult to apply broadly to marshes over regions and



time (Yellen et al., 2022). The health and resilience of marshes can be indicated across a greater spatial and temporal range by quantifying the extent of pool expansion and recovery. On marshes with greater pool cover and infrequent pool recovery, biogeochemical indices of health such as biomass, soil shear strength, and porewater are lower than those on marshes with greater vegetation coverage (Himmelstein et al., 2021). Sediment export due to erosion is greater on marshes with greater pool coverage and unvegetated soils (Ganju et al., 2013, 2020) because of the positive feedback driven by soil exposure to wind wave-driven erosion (Himmelstein et al., 2021; Schepers et al., 2020). If the rate of sea level rise exceeds accretion in the pool, it will not recover and instead, will continue to expand, forming mega-pools (Mariotti, 2016). Our results indicate that the feedback loop of erosion, particularly of mega-pools eroding without recovery, occurs on all the marshes we surveyed (Figure 3; Figure S3 in Supporting Information S1). Based on conclusions from previous studies, the continued pool expansion documented here further suggests a potential increase in sediment export from the marshes in this study because unvegetated soils are more vulnerable to wind and wave erosion (Ganju et al., 2013, 2020; Himmelstein et al., 2021). With more unvegetated areas exposed in mega-pools, marshes could become more susceptible to increased erosion and sediment loss. However, the potential increase in sediment export requires further study, as it is influenced by factors other than pool expansion, such as marsh sediment supply, and could have implications for marsh stability and coastal processes.

4.3. Restoration

While pools naturally occur (Redfield, 1972; K. R. Wilson et al., 2009), they also form due to hydrological interference from agriculture (Adamowicz et al., 2020; Burdick et al., 2020); our study has found that mega-pools which typically form on depressions resulting from agriculture have expanded since 2009 in part due to sea level rise and lasting effects of human activity on the marshes. Therefore, the results of our study support the need for restoration efforts with the intent to drain mega-pools and support re-vegetation. One restoration strategy could be implementing runnelling, which has been shown to be effective in draining mega-pools by reconnecting them to tidal channels with shallow ditches, allowing for re-vegetation (McKown et al., 2023), though runnelling is not a long-term solution for marsh resilience to sea level rise (Besterman et al., 2022). While in some cases, we observed a lack of re-vegetation following connection to drainage of some pools (Figure 6b), this does not suggest that runnelling will be ineffective because the mechanism by which the pools have been connected to drainage and the depth of drainage channels are different. Given recent and upcoming efforts of marsh restoration, it is important to document the changes of marsh pools to inform these and future efforts and quantify their efficacy as strategies to limit pool expansion. We need to reverse recent inundation trends to support resilient and healthy marshes to buffer storms, sequester carbon, and provide habitat for rare and endangered species. Future studies should focus on understanding the efficacy of these restoration methods and monitoring for any unintended consequences.

5. Conclusions

Salt marshes provide essential habitat for many endangered and threatened species, including birds. Marshes also act as natural buffers for coastal infrastructure, reducing the impact of storm surges and high tides. However, the expansion of pools, or standing water, on the marsh surface can diminish their capacity to provide these ecological benefits. We found an increase in pool area from 2009 to 2021 across all 12 marshes in this study, consistent with trends observed throughout the Northeastern US region. There was an increase in pool count per km² on all marshes except for the York River marsh, which still experienced an increase in area cover. Individual pool and perimeter pool area and count remained about the same, whereas increases in count and area were driven by megapool expansion. Mega-pools were larger on lower marsh elevations, whereas individual pools were larger on higher marsh elevations, suggesting that sea level rise may be prompting mega-pool expansion. Of the various ways pools can change over time, pools merging was the primary contributor of expansion, whereas pools that were present in 2009 and 2021 (remained singular), split into many pools, only existed in 2009, or only existed in 2021 contributed minimally to expansion and even decreased pool area. Goosefare Brook and Ogunquit River marshes experienced the greatest increase in pool cover, while the Webhannet marsh experienced the smallest increase over the 12 years. The Little River, Ogunquit, and Scarborough marshes have the greatest pool percent cover. Primary drivers of pool expansion could include rising sea levels and historic agricultural ditching, which continues to interfere with underlying marsh hydrology. Therefore, our results support the advocacy and need for marsh restoration projects to promote drainage and re-vegetation of mega-pools. In the coming decades, sea level



rise will significantly impact salt marshes worldwide and continued monitoring efforts are needed to ensure that marshes continue to provide numerous critical ecosystem services. Our work suggests that intervention is needed to reverse the expansion of pools and maintain marsh health.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

Data layers of pool polygons and python scripts have been made available at DUNE: DigitalUNE https://dune.une.edu/gis_data/1 (DeWater et al., 2024).

References

- Adamowicz, S. C., & Roman, C. T. (2005). New England salt marsh pools: A quantitative analysis of geomorphic and geographic features. Wetlands, 24(2), 279–288. https://doi.org/10.1672/4
- Adamowicz, S. C., Wilson, G., Burdick, D. M., Ferguson, W., & Hopping, R. (2020). Farmers in the marsh: Lessons from history and case studies for the future. Wetland Science and Practice, 37(3), 183–195.
- Alber, M., Swenson, E. M., Adamowicz, S. C., & Mendelssohn, I. A. (2008). Salt Marsh Dieback: An overview of recent events in the US. *Estuarine, Coastal and Shelf Science*, 80, 1–11. https://doi.org/10.1016/j.ecss.2008.08.009
- Alongi, D. M. (2020). Carbon balance in salt marsh and mangrove ecosystems: A global synthesis. Journal of Marine Science and Engineering, 8(10), 1–21. https://doi.org/10.3390/jmse8100767
- Berkowitz, J. F., VanZomeren, C. M., Piercy, C. D., & White, J. R. (2018). Evaluation of coastal wetland soil properties in a degrading marsh. *Estuarine, Coastal and Shelf Science*, 212, 311–317. https://doi.org/10.1016/j.ecss.2018.07.021
- Besterman, J. F., Jakuba, R. W., Ferguson, W., Brennan, D., Costa, J. E., & Deegan, L. A. (2022). Buying time with runnels: A climate adaptation tool for salt marshes. *Estuaries and Coasts*, 45(6), 1491–1501. https://doi.org/10.1007/s12237-021-01028-8
- Brewer, A., Burdick, D., Chmura, G., Christian, M., Colarusso, P., Crooks, S., et al. (2023). Region 1 The blue carbon reservoirs from Maine to Long Island (pp. 1–32). EPA.
- Burdick, D. M., Moore, G. E., Adamowicz, S. C., Wilson, G. M., & Peter, C. R. (2020). Mitigating the legacy effects of ditching in a New England salt marsh. *Estuaries and Coasts*, 43(7), 1672–1679. https://doi.org/10.1007/s12237-019-00656-5
- Burns, C. J., Alber, M., & Alexander, C. R. (2021). Historical changes in the vegetated area of salt marshes. *Estuaries and Coasts*, 44(1), 162–177. https://doi.org/10.1007/s12237-020-00781-6
- Burns, C. J., Alexander, C. R., & Alber, M. (2021). Assessing long-term trends in lateral salt-marsh shoreline change along a U.S. East Coast latitudinal gradient. Journal of Coastal Research, 37(2), 291–301. https://doi.org/10.2112/JCOASTRES-D-19-00043.1
- Cavorta, J. R., Johnston, M., Hopkinson, C., & Valentine, V. (2003). Patterns of sedimentation in a salt marsh-dominated estuary. *Biological Bulletin*, 205(2), 239–241. https://doi.org/10.2307/1543274
- Chambers, L. G., Steinmuller, H. E., & Breithaupt, J. L. (2019). Toward a mechanistic understanding of "peat collapse" and its potential contribution to coastal wetland loss. *Ecology*, 100(7), e02720. https://doi.org/10.1002/ecy.2720
- Corman, S. S., Roman, C. T., King, J. W., & Appleby, P. G. (2012). Salt marsh mosquito-control ditches: Sedimentation, landscape change, and restoration implications. *Journal of Coastal Research*, 28(4), 874–880. https://doi.org/10.2112/JCOASTRES-D-11-00012.1
- Crowell, M., Leatherman, S. P., & Buckley, M. K. (1991). Historical shoreline change: Error analysis and mapping accuracy. *Journal of Coastal Research*, 7(3), 839–852.
- DeLaune, R. D., Nyman, J. A., & Patrick, W. H. (1994). Peat collapse, ponding and wetland loss in a rapidly submerging coastal marsh. Journal of Coastal Research, 10(4), 1021–1030.
- DeWater, K., Kochtitzky, W., Ellis, R., & Merrill, P. M. (2024). Widespread expansion of salt marsh pools observed on Maine marshes since 2009 (data files) [Dataset]. *Geographic Information Systems Data Projects*, *1*. https://dune.une.edu/gis_data/1
- Erwin, R. M. (1996). Dependence of waterbirds and shorebirds on shallow-water habitats in the mid-Atlantic coastal region: An ecological profile and management recommendations. *Estuaries*, 19(2), 213–219. https://doi.org/10.2307/1352226
- Fernandez, I., Birkel, S., Schmitt, C., Simonson, J., Lyon, B., Pershing, A., et al. (2020). Maine's climate future 2020 update. University of Maine. Fletcher, C., Rooney, J., Barbee, M., Lim, S. C., & Richmond, B. (2003). Shoreline mapping and change analysis: Technical considerations and management implications. Journal of Coastal Research, 38, 106–124.
- Ganju, N. K., Defne, Z., & Fagherazzi, S. (2020). Are elevation and open-water conversion of salt marshes connected? *Geophysical Research Letters*, 47(3), e2019GL086703. https://doi.org/10.1029/2019GL086703
- Ganju, N. K., Zidzieko, N. J., & Kirwan, M. L. (2013). Inferring tidal wetland stability from channel sediment fluxes: Observation and a conceptual model. *Journal of Geophysical Research: Earth Surface*, 118, 2045–2058. https://doi.org/10.1002/jgrf.20143
- Gawler, S., & Cutko, A. (2015). *Natural Landscapes of Maine: A Guide to Natural Communities and Ecosystems*. Revised and updated edition 2018. Maine Natural Areas Program, Maine Department of Agriculture, Conservation and Forestry,.
- Goodman, J. E., Wood, M. E., & Gehrels, W. R. (2007). A 17-yr record of sediment accretion in the salt marshes of Maine (USA). *Marine Geology*, 242(1-3), 109–121. https://doi.org/10.1016/j.margeo.2006.09.017
- Hartig, E. K., Gorntiz, V., Kolker, A., Mushacke, F., & Fallon, D. (2002). Anthropogenic and climate-change impacts on salt marshes of Jamaica Bay, New York City. Wetlands, 22(1), 71–89. https://doi.org/10.1672/0277-5212(2002)022[0071:AACCIO]2.0.CO;2
- Himmelstein, J., Vincent, O. D., Temmerman, S., & Kirwan, M. L. (2021). Mechanisms of pond expansion in a rapidly submerging marsh. Frontiers in Marine Science, 8, 704768. https://doi.org/10.3389/fmars.2021.704768
- Kearney, M. S., Grace, R. E., & Stevenson, J. C. (1988). Marsh loss in Nanticoke Estuary, Chesapeake Bay. *Geographical Review*, 78(2), 205–220. https://doi.org/10.2307/214178
- Kelley, J. T., Belknap, D. F., Jacobson, G. L., & Jacobson, H. A. (1988). The morphology and origin of salt marshes along the glaciated coastline of Maine, USA. *Journal of Coastal Research*, 4(4), 649–666.

Acknowledgments

KD thanks the University of New England (UNE) for Summer Undergraduate Research Experience and Maine Space Grant Consortium funding for two summers (2023 and 2024) that supported this work. WK and KD thank UNE, the Maine Space Grant Consortium (EP-24-04), and the Maine Outdoor Heritage Fund for support. We thank Dr. Hanna Barannes for providing insights and feedback on our draft.



- Kirwan, M. L., & Guntenspergen, G. R. (2012). Feedbacks between inundation, root production, and shoot growth in a rapidly submerging brackish marsh. Journal of Ecology, 100(3), 764–770. https://doi.org/10.1111/j.1365-2745.2012.01957.x
- Kirwan, M. L., Guntenspergen, G. R., Alpaos, A. D., Morris, J. T., Mudd, S. M., & Temmerman, S. (2010). Limits on the adaptability of coastal marshes to rising sea level. *Geophysical Research Letters*, 37(23), L23401. https://doi.org/10.1029/2010GL045489
- Kirwan, M. L., & Mudd, S. M. (2012). Response of salt-marsh carbon accumulation to climate change. Nature, 489(7417), 550–554. https://doi. org/10.1038/nature11440
- Kirwan, M. L., & Murray, B. (2007). A coupled geomorphic and ecological model of tidal marsh evolution. PNAS, 104(15), 6118–6122. https:// doi.org/10.1073/pnas.0700958104
- Luk, S., Eagle, M. J., Mariotti, G., Gosselin, K., Sanderman, J., & Spivak, A. C. (2023). Peat decomposition and erosion contribute to pond deepening in a temperate salt marsh. *Journal of Geophysical Research: Biogeosciences*, 128(2), e2022JG007063. https://doi.org/10.1029/ 2022JG007063
- Maine Climate Council. (2020). Scientific Assessment of Climate Change and Its Effects in Maine. A Report by the Scientific and Technical Subcommittee (STS) of the Maine Climate Council (MCC) (p. 370). Augusta.
- Mariotti, G. (2016). Revisiting salt marsh resilience to sea level rise: Are ponds responsible for permanent land loss? *Journal of Geophysical Research: Earth Surface*, 121(7), 1391–1407. https://doi.org/10.1002/2016JF003900
- Mariotti, G., & Fagherazzi, S. (2013). Critical width of tidal flats triggers marsh collapse in the absence of sea-level rise. Earth, Atmospheric, and Planetary Sciences, 110(14), 5353–5356. https://doi.org/10.1073/pnas.1219600110
- Mariotti, G., Spivk, A. C., Luk, S. Y., Ceccherini, G., Tyrrell, M., & Gonneea, M. E. (2020). Modeling the spatial dynamics of marsh ponds in New England salt marshes. *Geomorphology*, 365, 107262. https://doi.org/10.1016/j.geomorph.2020.107262
- McKown, J. G., Burdick, D. M., Moore, G. E., Peter, C. R., Payne, A. R., & Gibson, J. L. (2023). Runnels reverse mega-pool expansion and improve marsh resiliency in the Great Marsh, Massachusetts (USA). Wetlands, 43(4), 35. https://doi.org/10.1007/s13157-023-01683-6
- Morris, J. T., Sundareshwar, P. V., Nietch, C. T., Kjerfve, B., & Cahoon, D. R. (2002). Responses of coastal wetlands to rising sea level. *Ecology*, 83(10), 2869–2877. https://doi.org/10.1890/0012-9658(2002)083[2869:ROCWTR]2.0.CO;2
- OCM Partners. (2024). 2020 USGS Lidar: South Coastal Maine (QL2) from 2010-06-15 to 2010-08-15. NOAA National Centers for Environmental Information. Retrieved from https://www.fisheries.noaa.gov/inport/item/65827
- Odum, W. E. (1988). Comparative ecology of tidal freshwater and salt marshes. Annual Review of Ecology and Systematics, 19(1), 147–176. https://doi.org/10.1146/annurev.ecolsys.19.1.147
- Peck, E. K., Walker, J. E., Woodruff, J. D., & Ganju, N. K. (2024). Linear Ditches of Northeastern U.S. Coastal Marshes from Maine to Virginia Derived from 2023 2D Aerial Imagery Basemap. U.S. Geological Survey Data Release. https://doi.org/10.5066/P13WF8ZT
- Pethick, J. S. (1974). The distribution of salt pans on tidal salt marshes. *Journal of Biogeography*, *1*(1), 57–62. https://doi.org/10.2307/3038068 Raposa, K. B., Cole Ekberg, M. L., Burdick, D. M., Ernst, N. T., & Adamowicz, S. C. (2017). Elevation change and the vulnerability of Rhode
- Island (USA) salt marshes to sea-level rise. *Regional Environmental Change*, *17*(2), 389–397. https://doi.org/10.1007/s10113-016-1020-5 Redfield, A. C. (1972). Development of a New England salt marsh. *Ecological Monographs*, *42*(2), 201–237. https://doi.org/10.2307/1942263
- Schepers, L., Brennand, P., Kirwan, M. L., Guntenspergen, G. R., & Temmerman, S. (2020). Coastal marsh degradation into ponds induces irreversible elevation loss relative to sea level in a microtidal system. *Geophysical Research Letters*, 47(18), e2020GL089121. https://doi.org/ 10.1029/2020GL089121
- Sebold, K. R. (1998). "The low green prairies of the sea": Economic usage and cultural construction of the Gulf of Maine salt marshes. Doctoral dissertation (pp. 1–315). University of Maine.
- Smith, D. C., & Bridges, A. E. (1982). Salt marsh dykes (dikes) as a factor in eastern Maine agriculture. Maine History, 21(4), 219–226.
- Smith, D. C., Konrad, V., Koulouris, H., Hawes, E., & Borns, H. W., Jr. (1989). Salt marshes as a factor in the agriculture of northeastern North America. Agricultural History, 63(2), 270–294.
- Smith, J. A. M., Adamowicz, S. C., Wilson, G. M., & Rochlin, I. (2022). Waffle' pools in ditched salt marshes: Assessment, potential causes, and management. Wetlands Ecology and Management, 30(5), 1081–1097. https://doi.org/10.1007/s11273-021-09835-3
- Smith, S. M., Tyrrell, M., Medeiros, K., Bayley, H., Fox, S., Adams, M., et al. (2017). Hypsometry of Cape Cod salt marshes (Massachusetts, U.S. A.) and predictions of marsh vegetation responses to sea-level rise. *Journal of Coastal Research*, 33(3), 537–547. https://doi.org/10.2112/ JCOASTRESD1500153.1
- Temmerman, S., De Vries, M. B., & Bouma, T. J. (2012). Coastal marsh die-off and reduced attenuation of coastal floods: A model analysis. Global and Planetary Change, 92–93, 267–274. https://doi.org/10.1016/j.gloplacha.2012.06.001
- Vincent, R. E., Burdick, D. M., & Dionne, M. (2013). Ditching and ditch-plugging in New England salt marshes: Effects on hydrology, elevation, and soil characteristics. *Estuaries and Coasts*, 37(3), 354–368. https://doi.org/10.1007/s12237-012-9583-y
- Vincent, R. E., Burdick, D. M., & Dionne, M. (2014). Ditching and ditch-plugging in New England salt marshes: Effects on plant communities and self-maintenance. *Estuaries and Coasts*, 36(2), 610–625. https://doi.org/10.1007/s12237-013-9671-7
- Watson, E. B., Raposa, K. B., Carey, J. C., Wigand, C., & Warren, R. S. (2017). Anthropocene survival of southern New England's salt marshes. *Estuaries and Coasts*, 40(3), 617–625. https://doi.org/10.1007/s12236-016-0166-1
- Watson, E. B., Wigand, C., Davey, E. W., Andrews, H. M., Bishop, J., & Raposa, K. B. (2016). Wetland loss patterns and inundation-productivity relationships prognosticate widespread salt marsh loss for southern New England. *Estuaries and Coasts*, 40(3), 662–681. https://doi.org/10. 1007/s12237-016-0069-1
- Wigand, C., Ardito, T., Chaffee, C., Ferguson, W., Paton, S., Raposa, K., et al. (2017). A climate change adaptation strategy for management of coastal marsh systems. *Estuaries and Coasts*, 40(3), 682–693. https://doi.org/10.1007/s12237-015-0003-y
- Wilson, C. A., Hughes, Z. J., FitzGerald, D. M., Hopkinson, C. S., Valentine, V., & Kolker, A. S. (2014). Saltmarsh pool and tidal creek morphodynamics: Dynamic equilibrium of northern latitude saltmarshes? *Geomorphology*, 213, 99–115. https://doi.org/10.1016/j.geomorph. 2014.01.002
- Wilson, K. R., Kelley, J. T., Croitoru, A., Dionne, M., Belknap, D., & Steneck, R. (2009). Stratigraphic and ecophysical characterizations of salt pools: Dynamic landforms of the Webhannet Salt Marsh, Wells, ME, USA. *Estuaries and Coasts*, 32(5), 855–870. https://doi.org/10.1007/ s12237-009-9203-7
- Wilson, K. R., Kelley, J. T., Tanner, B. R., & Belknap, D. F. (2010). Probing the origins and stratigraphic signature of salt pools from northtemperate marshes in Maine, U.S.A. Journal of Coastal Research, 26(6), 1007–1026. https://doi.org/10.2112/JCOASTRES-D-10-00007
- Yellen, B., Woodruff, J. D., Baranes, H. E., Engelhart, S. E., Geyer, W. R., Randall, N., & Griswold, F. R. (2022). Salt marsh response to inlet switch-induced increases in tidal inundation. *Journal of Geophysical Research: Earth Surface*, 128(1), e2022JF006815. https://doi.org/10. 1029/2022JF006815